Author's response to reviews

Title: The descriptive epidemiology of delirium symptoms in a large population-based cohort study: results from the Medical Research Council Cognitive Function and Ageing Study (MRC CFAS)

Authors:

Daniel H.J. Davis (dhjd2@cam.ac.uk)
Linda Barnes (leb22@medschl.cam.ac.uk)
Blossom Stephan (blossom.stephan@newcastle.ac.uk)
Alasdair M.J. MacLullich (a.maclullich@ed.ac.uk)
David Meagher (david.meagher@ul.ie)
John Copeland (jrmcop@btinternet.com)
Fiona Matthews (fiona.matthews@mrc-bsu.cam.ac.uk)
Carol Brayne (cb105@medschl.cam.ac.uk)

Version: 5
Date: 8 July 2014

Author's response to reviews:

LREC information added in text:

Each center had ethical approval from the Local Research Ethics Committee (Cambridge: North West Anglia Health Authority Local Research Ethics Committee (Peterborough); Huntingdon Local Research Ethics Committee; Cambridge Local Research Ethics Committee. Gwynedd: Gwynedd Hospitals NHS Trust – NWHA Research Ethics Committee (West). Newcastle: Newcastle & North Tyneside Health Authority – Joint Ethics Committee; Northumberland and Tyne & Wear Health Authority – Local Research Ethics Committee. Nottingham: QMC NHS Trust Ethics Committee; Nottingham University Medical School Ethical Committee; City Hospital Ethics Committee. Oxford: Oxfordshire Health Authority: Central Oxford Research Ethics Committee).